Author's response to reviews

Title: Policy Maker and Provider Knowledge, Attitudes regarding the Provision of Emergency Contraceptive Pills within Lao PDR

Authors:

Vanphanom Sychareun (vsychareun@gmail.com)
Keokedthong Phongsavan (keota_63@hotmail.com)
Visanou Hansana (visanou65@yahoo.com)
Alongkone Phengsavanh (ogalk@yahoo.com)

Version: 2 Date: 1 April 2010

Author's response to reviews:

Please find enclosed a revision of the paper entitled “Policy Policy Maker and Provider Knowledge, Attitudes regarding the Provision of Emergency Contraceptive Pills within Lao PDR” to you for consideration for publication in your journal. I did answered to the questions of the three reviewers.

All authors have read this final manuscript and have given their approval to be submitted in its present form. For the authorship, all authors have contributed significantly, and that all authors are aware that they are being listed, have been sent and approved the manuscript and accept responsibility for its content.

As the corresponding author, my contact details are shown on the cover page of the manuscript. I hope that this proposal should be forwarded into your consideration. If you have any questions to the revision, please send me the feedback.

I am looking to hearing from you soon.

Yours sincerely,

Vanphanom Sychareun